Author’s response to reviews

Title: Relationship between students’ perceptions of the adequacy of M1 and M2 curricula and their performance on USMLE Step 1 examination

Authors:

Mohammed Khalil (khalilmk@greenvillemmed.sc.edu)

William Wright (WRIGH288@greenvillemmed.sc.edu)

Kelsey Spearman (kelseyag@email.sc.edu)

Amber Gaspard (agaspard@email.sc.edu)

Version: 2 Date: 03 Sep 2019

Author’s response to reviews:

Editor Comment 1: The final four lines to the “Methods: Data Analysis” section have been revised based on the reviewer comments provided.

Editor Comment 2: Abbreviations have been added to the manuscript in the appropriate location on page 14.

Editor Comment 3: The statement in the “ethical approval and consent to participate” has been updated to reflect student consent was written.